\def\GW{/GW}
\def\visibility{true}
\input{\GW/root/GW4reports/GWlatexBeamerHead.tex}


\AtBeginSection[]
{
  \begin{frame}
    \frametitle{Table of Contents}
    \tableofcontents[currentsection]
  \end{frame}
}

\begin{document}

\title{Tokamak GOLEM}
\author{Vojt\v{e}ch Svoboda \\ Přednáška pro FS (C\#19 kompenzace)}

\slide{Title}{\titlepage}


\GWis{Staff/Employees/VoSv/Introduction}
\GWin{Theory/Fusion/FusionIntro/ThermalPowerPlant}
\SlunceElektr
\GWisCz{Theory/Fusion/FusionIntro/HarnessEnergy}
\FuzniElektr
%\GWisCz{Theory/Fusion/FusionIntro/PragueEnergyDemands}
%\GWisCz{Theory/Fusion/FusionIntro/OtherProjects/Sun.un/ProtonProtonChain}
%\GWisCz{Theory/Fusion/FusionIntro/DeltaMass}
%\GWisCz{Theory/Fusion/FusionIntro/FusionReaction}
%\GWis{Theory/Fusion/FusionIntro/IAEA_FusionWater}
%\GWisCz{Theory/Fusion/FusionIntro/FusionWaste}
%\GWisCz{Theory/Fusion/FusionIntro/SafetyIssues}
\GWisCz{Theory/Fusion/FusionIntro/Hbomb}

\GWisCz{Theory/Fusion/FusionIntro/FusionTechnology}
\GWisCz{Theory/TokamakTechnology/Heating/Overview}
\GWis{Theory/Fusion/FusionIntro/ThreeWaysToConfinePlasma/cz}
\GWisCz{Theory/Fusion/FusionIntro/InertialFusion}
\GWisCz{Theory/Fusion/FusionIntro/MagneticBottle}

\def\GWinclMovie{2}
\GWisCz{Tools/VirtualModels/TatOko_OG/SunInTheLabMovie}
\GWisCz{Tokamak/Plasma/PlasmaRingMovie}
\GWisCz{Tokamak/VacuumVessel/ChamberMovie}
\GWisCz{Tokamak/ToroidalMagneticFieldCoils/MovieSlide}
\GWisCz{Tokamak/Transformer/MovieSlide}
\GWisCz{Tokamak/Overview/MovieSlide}

\GWis{root/GW4reports/ConJoints/Openings/Milestones}
\GWisCz{Theory/Fusion/FusionIntro/OtherProjects/JET.eu/WorldRecord}
%\GWisCz{Theory/Fusion/FusionIntro/FusionPowerPlant}
\GWisCz{Theory/Fusion/FusionIntro/OtherProjects/ITER.world}
\GWisCz{Theory/Fusion/FusionIntro/OtherProjects/Compass.cz}


\GWisCz{root/GW4reports/ConJoints/Intros/EducationKeyToSuccess}
\GWisp{Tokamak/BasicDescription/Introduction}{\BasePhotoCz}
\GWis{History/GolemCompassHistory}
\GWisp{root/GW4reports/ConJoints/Openings/GolemStory/GOLEM}{\PureGolem}
%\GWis{Tokamak/BasicDescription/TokamakGOLEMconcept}
\GWis{Tokamak/Plasma/ScenarioToDischarge/Basic}
\GWinput{root/GW4reports/ConJoints/Specifics/LetUsMakeDischarge/latexsrc.page}\discharge


\GWisCz{Tokamak/ExperimentalSetup/GlobalSetup}

\GWisp{Chronicle/Inventory/HandsOn}{\KFpract}
\GWisCz{Chronicle/Inventory/Remotes/2019}
\GWisCz{root/GW4reports/ConJoints/Closings/Fee}
%\GWis{root/GW4reports/ConJoints/Closings/DischargesAvailable/0719ColoursOfOstrava}
\GWisp{root/GW4reports/ConJoints/Openings/GolemStory/GOLEM}{\PureGolem}

\GWisCz{root/GW4reports/ConJoints/Closings/ThankYouThreeStagesOfTokamakGOLEM}
\GWis{root/GW4reports/ConJoints/Intros/WikiOldestTokamak}
\GWis{Production/Acknowledgment/2018}

\nocite{TED:BreneBrown}
\nocite{FusenEngDes11}
\nocite{BrotancePhD}
\nocite{wwwgolem}
\nocite{Wesson:2004}
\begin{frame}[allowframebreaks] \frametitle{References}
\bibliographystyle{unsrt}
\bibliography{/GW/root/bibtex/golem,/GW/root/bibtex/others}
\end{frame}


\GWis{Chronicle/TimeLine/0160TM1}
\GWis{Chronicle/TimeLine/0101CASTOR}
\GWis{Chronicle/TimeLine/1207Transport}
\GWis{Chronicle/TimeLine/0709FirstPlasma}
\GWis{Chronicle/TimeLine/0909TokamakAndTokamak}
\GWis{Chronicle/SpecialGuests/1111Barish}
\GWis{Chronicle/Misc/0516YoungestOperatorAdam}
\GWis{Chronicle/SpecialGuests/0916BB}
\GWis{Events/ABCday/Model}
\GWis{Chronicle/Tokamaks/FTTF@Marianska/10}
\GWis{Chronicle/Tokamaks/FTTF@Marianska/11}
\GWis{Chronicle/Tokamaks/FTTF@Marianska/16}
\GWis{Chronicle/Tokamaks/FTTF@Marianska/17}
\GWis{Chronicle/Smiles/1015TrojskyKun20k}
\GWis{Chronicle/TimeLine/1117MappingNnJ}
\GWis{Chronicle/Statistics/OveralPlasmaLife/1112}
\GWisp{Chronicle/Smiles/0916Reholnice}\TheGod


\end{document}



